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CERES Processing Oversight Board Meeting Notes 

April 29, 2015 

 

Meeting began at 1:00 p.m. in B2102 R114. 

Meeting Notes  

There were no comments or questions on the notes from the last meeting.   

Announcements 

 CERES Science Team Meeting will be held May 5 – 7 in the Pearl Young Theater. 

 

SCCR Discussion (SCCRs that have been submitted or updated since the last meeting 

There was no SCCR activity in the past two weeks. 

  

CATALYST Status  

SSL RT ticket was submitted.  This will enable https for the site. 

 

CATALYST schedule will be updated next week to reflect new delivery date. 

 

Screenshots of the Operator’s Console will be provided for Jonathan’s Science Team Meeting 

presentation. 

    

PR Tool Status 

Science dev version available for stream review. 

 

PR Tool delivery  needs to be timed with delivery of CATALYST 2.0.  Open SSL and 

CATALYST ready flag set to “true” will be in this delivery. 

 

Production Processing Status 

Updates to posted report: 

Table 2 – Complete through March 31. 2015. 

Table 7 – Terra complete through July 2011, Aqua complete through February 2011. 

 

Walt reported that Raja delivered GGEO data needed yesterday. 

 

System Status   

There will be a follow-up meeting discussing the P7 blades after the CPOBM. 

  

Status of Open SCCRs 

1059 – In PPE testing. Should be promoted soon.  

1023 – Waiting for a decision from Dave Doelling on whether PGE CER10.0P5 should be P6 

only.  Will need to deliver an updated script for P6-only. 

1038 – Waiting for PRs.  1059 has to be promoted before PRs can be written. 

1024 – Finished running Test Plan test cases.  Using email for operational testing.  Will need 

PRs for additional testing.   

1017 – Finished running Test Plan test cases.  Using email for operational testing.  Will need 

PRs for additional testing.  
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1025 – Finished running Test Plan test cases.  Using email for operational testing.  Will need 

PRs for additional testing. 

1069 –Waiting for Dave Doelling to approve.  Delivery delayed by a week. 

1047 – Validating ValRx runs.  Need to have a discussion with Denise about upcoming IES 

changes before this can be promoted. 

1064 – In Vertley’s queue.  

1058 -  Delivering after Edition3/Edition4 runs through 2014. 

1049 – Status changed to “Long Term”.  Still working on algorithms with Norman and Pat. 

1050 – Status changed to “Long Term”.  Still working on algorithms with Norman and Pat.  

1052 – Delivery date dependent on EDOS finishing testing. 

1060 – No updates.   

1061 – No updates. 

1065 – New delivery date will be determined soon. 

1066 – New delivery date will be determined soon. 

1067 – New delivery date will be determined soon. 

1068 – New delivery date will be determined soon. 

1071 – No updates. 

 

 

Meeting ended at 1:38 p. m. 


